Author's response to reviews

Title: Haploinsufficiency for BRCA1 is associated with normal levels of DNA nucleotide excision repair in breast tissue and blood lymphocytes

Authors:

Jean J Latimer (latimerj@pitt.edu)
Wendy S Rubinstein (wrubinstein@enh.edu)
Jennifer M Johnson (rsijj@mwri.magee.edu)
Amal Kanbour-Shakir (shakir@pitt.edu)
Victor G Vogel (vvogel@mail.magee.edu)
Stephen G Grant (sgg@pitt.edu)

Version: 2 Date: 14 February 2005

Author's response to reviews:

To the editors,

The requested change (addition of a "Methods" section in the Abstract) has been made.

J. Latimer